Author’s response to reviews

Title: Prevalence and clinical impact of Streptococcus pneumoniae nasopharyngeal carriage in solid organ transplant recipients

Authors:

Cristina Roca-Oporto (cristinaroca85@gmail.com)
Tania Cebrero-Cangueiro (tany5_91@hotmail.com)
María Luisa Gil-Marqués (malu_tss@hotmail.com)
Gema Labrador-Herrera (gemalabrador@hotmail.com)
Younes Smani (ysmani-ibis@us.es)
Francisco M. González-Roncero (roncero.roni@gmail.com)
Luis Miguel Marín-Gómez (lmarin@us.es)
Jerónimo Pachón (pachon@us.es)
Maria Eugenia Pachón-Ibáñez (mpachon-ibis@us.es)
Elisa Cordero (elisacorderom@gmail.com)

Version: 3 Date: 24 Jul 2019

Author’s response to reviews:

BMC Infectious Diseases
July 24, 2019

Dear Dr. Oliver Cocks,

Please find the revised version of the manuscript "Prevalence and clinical impact of Streptococcus pneumoniae nasopharyngeal carriage in solid organ transplant recipients" submitted to BMC Infectious Diseases following the Editor’ comments.

We have included all the necessary corrections detailed by the Editor.

We hope that this revised version of the manuscript will fulfil the requirements of BMC Infectious Diseases for publication.
Kindest regards,

Mª Eugenia Pachón Ibáñez, BS.c, Ph.D.
Clinical Unit of Infectious Diseases, Microbiology, and Preventive Medicine
Infectious Diseases Research Group
Institute of Biomedicine of Seville (IBiS)
University of Seville/CSIC/University Hospital Virgen del Rocío
Seville, Spain
Phone: +34 955923100